Author's response to reviews

Title: A Long-term follow-up of the imatinib mesylate treatment for the patients with recurrent gastrointestinal stromal tumor: the liver metastasis and the outcome

Authors:

Jiang Zhu (zhujiang1@medmail.com.cn)
Yu Yang (yangyuuflying@hotmail.com)
Lin Zhou (drzhoulin@gmail.com)
Ming Jiang (mingminyu@gmail.com)
Mei Hou (hxhoumei@gmail.com)

Version: 4 Date: 8 May 2010

Author's response to reviews:

Dear editor:

Thank you for the acceptance of our manuscript. We had checked and changed the format carefully. We hope this final revision will meet your request.

Here, we are prepared to response to your comments point-to-point.

Major revisions

References: Please make sure all references are cited in numerical order. We notice reference 21 is not cited within the main text.
Answer: Reference 21 is not proper for this revision. It had been deleted by the authors.

Minor revisions

Title: please use sentence case in the title, both in the manuscript file and the submission system. Capitalise only the first word and proper nouns, and do not include full stops/periods.
Answer: Checked.

Title page: Please remove the running head from the title page.
Answer: The running head was removed.

Email address: please include email addresses for all the authors on the title page, using the following format:……
Answer: E-mail addresses for all the authors are provided in the title page.

Abstract: Please remove the key words from the abstract.
Answer: The key words were removed.
Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.
Answer: The highlighting and tracking was removed.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
Answer: Checked.

Thank you very much.

Yours sincerely,

Dr. Jiang Zhu & Dr. Mei Hou